



Using Generative Artificial Intelligence Tools to Explain and Enhance Experiential Learning for Authentic Assessment

David Ernesto Salinas-Navarro ¹,*^(D), Eliseo Vilalta-Perdomo ¹^(D), Rosario Michel-Villarreal ²^(D) and Luis Montesinos ³,*^(D)

- ¹ Community Resilience and Sustainability Education Lab (CoRSEL), Aston University, Birmingham B4 7ET, UK; e.vilaltaperdomo@aston.ac.uk
- ² Sustainability Research Institute, School of Earth and Environment, University of Leeds, Leeds LS2 9JT, UK; r.michel-villarreal@leeds.ac.uk
- ³ Institute of Advanced Materials for Sustainable Manufacturing, Tecnologico de Monterrey, Mexico City 14380, Mexico
- * Correspondence: d.salinas-navarro@aston.ac.uk (D.E.S.-N.); lmontesinos@tec.mx (L.M.)

Abstract: The emergence of generative artificial intelligence (GenAI) requires innovative educational environments to leverage this technology effectively to address concerns like academic integrity, plagiarism, and others. Additionally, higher education needs effective pedagogies to achieve intended learning outcomes. This emphasizes the need to redesign active learning experiences in the GenAI era. Authentic assessment and experiential learning are two possible meaningful alternatives in this context. Accordingly, this article investigates how GenAI can enhance teaching and learning by constructively addressing study situations beyond conventional learning approaches and cultivating high-order skills and knowledge acquisition. This study employs thing ethnography to examine GenAI tools' integration with authentic assessment and experiential learning and explore implementation alternatives. The results reveal insights into creating human-centered and GenAI-enhanced learning experiences within a constructive alignment. Specific examples are also provided to guide their implementation. Our contributions extend beyond the traditional use of GenAI tools as mere agents-to-write or agents-to-answer questions to become agents-to-support experiential learning for authentic assessment. These findings underscore the transformative role of GenAI tools in enhancing teaching and learning efficacy and effectiveness. The limitations in treating GenAI tools as subjects in thing ethnography are acknowledged, with potential for future implementation evaluation.

Keywords: experiential learning; authentic assessment; constructive alignment; generative artificial intelligence; educational innovation; higher education

1. Introduction

This research-to-practice work refers to using generative artificial intelligence (GenAI) in experiential learning activities for authentic assessment in higher education (HE). This work arises from the current concerns about using GenAI tools in educational activities; the potential risks for academic integrity, intellectual property, and plagiarism; and the adverse effects on students' learning, skill development, and knowledge acquisition.

GenAI is revolutionizing different aspects of daily life, as reported in news media headlines, including professional practice, education, and science, by generating various texts, images, audio, algorithms, or combinations of them [1]. However, GenAI has raised concerns about developing trust in these artifacts, controlling their creation and managing their adoption [2,3].

GenAI can be defined as "the field of science which studies the (fully) automated construction of intelligence" [4]. GenAI involves machine learning and pre-trained large language models based on a large corpus of text data, learning grammar, vocabulary, and various linguistic elements to later generate coherent and contextually relevant human-like



Citation: Salinas-Navarro, D.E.; Vilalta-Perdomo, E.; Michel-Villarreal, R.; Montesinos, L. Using Generative Artificial Intelligence Tools to Explain and Enhance Experiential Learning for Authentic Assessment. *Educ. Sci.* 2024, *14*, 83. https://doi.org/ 10.3390/educsci14010083

Academic Editor: Diego Vergara

Received: 16 December 2023 Revised: 5 January 2024 Accepted: 10 January 2024 Published: 12 January 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). content in response to the complex prompts it receives. Among the specific application tools of GenAI, ChatGPT 3.5 (Chat Generative Pre-Trained Transformer) is the most famous because it was one of the first tools that were made free and easily accessible online [3]. Nonetheless, an improved paid version (Chat GPT 4) is now available.

ChatGPT, for instance, can be used in education to write assignments, articles, and presentation slides; elaborate and answer exams; or solve coursework problems. These possibilities raise concerns given its intellectual transformative power and limitations concerning its information sources, insufficient training, false responses, misleading information, and the potential generation of spam, hate speech, and other harmful associations that might be implicit in its contained data [3]. However, other academic concerns also point to the scientific reliability and ethical implications of knowledge homogenization, rethinking learning outcome assessments and (higher-order) thinking processes [5–7].

Moreover, the use of GenAI should take some considerations, for pedagogical reasons, to prioritize human agency and responsible use [7]. These considerations include contributing to humankind's needs and learning effectiveness; supporting intrinsic (learning) motivation; the technology that humans control; the learning purpose and the learner's profile; and promoting human interactive engagement, higher-order thinking, and human accountability usage and impact. Hence, appropriate GenAI interactions should consider a definition of proper knowledge domain applications, clear outcomes, suitable tools and comparative advantages, users' requirements, human pedagogical methods, and ethical risks.

Some recent research results show that ChatGPT can provide answers in exams that exceed the mean responses of students, which poses a significant challenge to traditional assessment methods in HE [8]. These findings highlight the need to redesign curricula and methods of assessment through, for instance, reintroducing invigilated, in-person assessments, augmenting experience with chatbots, and increasing the prevalence of practical projects that artificial intelligence struggles to replicate well.

Nevertheless, GenAI tools can also enhance student learning by aiding in preparing for and writing assignments and improving their quality and narratives [9]. Using this technology for personalized, self-directed, and adaptive learning and ubiquitous on-demand support is also a potential gain. For example, GenAI tools can provide information and customized learning plans, generate feedback, and offer complementary learning resources to students at any time [10,11].

Previous works on information technology point to its diverse applications to enhance learning effectiveness and efficacy, for instance, by using artificial neural networks to predict academic performance [12]; web-enabled self-regulated learning [13]; games, mixed reality, social media, and other tools for ICT-supported pedagogical practices [14]; and simulation games [15]. However, the use of AI opens new learning enhancement opportunities.

An alternative to using GenAI in education, beyond exams and assignment writing, is building supportive and engaging learning environments that complement traditional pedagogical methods. This proposition can provide a dynamic and interactive platform to foster knowledge acquisition and acknowledge the existing concerns on plagiarism and academic integrity [8,16]. However, this perspective requires effective teacher leadership to guide the adequate use of GenAI tools. Additionally, GenAI can ignite the creation of innovative *authentic assessments* and irreplicable learning experiences by asking students to demonstrate comprehension and apply knowledge to complex and fictitious cases [16]. Authentic assessment refers to examining student performance on worthy intellectual tasks [17].

This learning environment refers to *experiential learning* that goes beyond simple memorization and fosters a deeper understanding of academic subjects through reflective and practical activities [8]. Experiential learning emphasizes what students must do to construct their knowledge and achieve their intended learning outcomes [18]. Therefore, assessments that evaluate higher-level cognitive skills like analysis, creation, and evaluation

can help engage students in meaningful learning experiences while making it more difficult for GenAI tools to deal with them.

Accordingly, GenAI tools can support teachers (or academics) as agents-to-thinkwith through a constructionist view, fostering more interactive and engaging learning experiences and promoting more profound understanding, critical thinking, and handson activities in students' fields [19]. Hence, learning activities and assessment methods should promote higher-level learning, whether teaching in-person or online, that can make a significant impact on students' learning outcomes. With this in mind, there is a pending task for instructional designers and teachers to develop authentic assessment and experiential learning practices using GenAI tools to support students' learning effectively [20].

Accordingly, this work aims to explore alternatives, as actionable recommendations, to carry out experiential learning-based activities and authentic assessments that integrate GenAI tools in HE. By doing so, we intend to answer the following research questions (RQ):

- 1. (RQ1) What is the interplay between GenAI tools and experiential learning for authentic assessment?
- 2. (RQ2) What alternatives can be identified for including GenAI tools in learning activities while concurrently considering experiential learning and authentic assessment?

These questions entail (i) a clarification of the notions of authentic assessment and experiential learning using GenAI tools, (ii) establishing a relationship between GenAI tools and experiential learning for authentic assessment, and (iii) investigating alternatives for the use of GenAI tools in specific learning activities.

A working hypothesis is proposed to guide the research process comprising the central notions supporting this work:

Incorporating GenAI tools into learning activities while concurrently considering experiential learning and authentic assessment can help support students' learning effectively.

Following these ideas, this article unfolds in five additional sections. Section 2 reviews the primary conceptual constructs supporting this work: experiential learning, authentic assessment, and constructive alignment. Section 3 covers the methodology of this work by using a *thing ethnography* approach that integrates the conceptual constructs guiding this work to interview GenAI tools. Thing ethnography considers things, which are not objects, but subjects that possess a non-human worldview or perspective to unveil novel insights in the research [21–23]. Section 4 summarizes the results, and Section 5 discusses the results, findings, limitations, and future work. Finally, Section 6 refers to the conclusions of this work amid the research aim and the research questions.

2. Background

Given the existing literature on the use of GenAI tools in education and their contribution to the effective development of student learning outcomes, the concepts of experiential learning and authentic assessment need clarification and integration as the primary conceptual constructs supporting this work.

2.1. Experiential Learning

HE demands pedagogical approaches that consider real-world situations to gain relevant learning and build new capabilities in students for their future professional careers [24]. Moreover, these approaches should allow for long-lasting learning in diverse environments and from multiple perspectives [25].

Accordingly, experiential learning is considered a more effective alternative than any other educational approach for high-impact education as it enhances students' motivation to construct meaningful learning [26]. This type of learning might be seen in terms of experience-based, reflective, and problem-solving activities.

This type of learning requires moving from a knowledge-broadcasting kind of teaching, where students passively sit and listen, to a constructivist alternative in which students learn by thoughtfully executing tasks while being immersed in a meaningful situation. There is the assumption that by providing students with experiential learning, they will

have the motivation and engagement to achieve their expected learning results [27]. Hence, experiential learning turns into a first-hand alternative to support active learning.

Experiential learning is widely acknowledged as part of a continuous meaning-making process in specific contexts, whereby students develop an interest in and recognize learning relevance through personal and environmental experiences [28]. Kolb's experiential learning cycle, which involves four stages, including concrete experience (CE), reflective observation (RO), abstract conceptualization (AC), and active experimentation (AE), is the backbone of experiential learning [18]. Each cycle stage depends on its predecessor and follows a continuous logical pattern step by step. CE refers to a new experience or situation that triggers a stimulus to actively engage in a task rather than merely reading or watching. RO is about reflecting on the new experience and recognizing any possible discrepancies and gaps between the learner's understanding and the experience. AC concerns new ideas or modified thoughts coming out from the reflection. It also includes interpreting and updating experiences from new knowledge. Finally, AE refers to what the learner applies to the outer world. It is also known as the testing stage to apply conclusions to new experiences. Thus, learning comprises intellectual activities that guide learners' actions, whereas practical activities and tasks provide feedback for conceptual knowledge in a specific context.

Approaching learning from this perspective helps to systematize reflective practice for hands-on, meaningful activities in situated real-world scenarios to develop students' learning outcomes [29,30].

2.2. Authentic Assessment

An assessment is authentic when student performance is directly examined on worthy intellectual tasks rather than indirect, proxy, or simplistic substitutes from which valid inferences are made [17]. Authentic assessments enable learners to address realistic tasks, ambiguity, or actual intellectual challenges to judge, clarify, and take purposeful action toward mastery in a learning situation [31]. Therefore, student understanding is seen as the ability to explore, criticize, or extend theories and assumptions, and knowledge is thus displayed as reflective know-how. All authentic assessments are performance assessments because they require students to construct extended responses effectively, perform a task, or produce a product [32].

Consequently, in this view, authentic assessment allows for more sophisticated and effective ways to use knowledge, for instance, in contextualized problem-solving and decision-making situations to develop complex and critical thinking [17,31].

Additionally, authentic assessments use authentic performance standards that are inherent to successful performance (of what students can do), including multifaceted scoring systems disaggregated for judging learning achievements rather than relying on scoring tests [17,31]. Accordingly, authentic assessments should, for instance, require students to mirror the priorities and challenges found in the activities of academic disciplines and professional practice or simulate real-world tests of ability, among others.

Authentic assessment comprises crucial principles in pedagogical design [33,34]. It incorporates realism by presenting situations or scenarios of real-life or professional contexts, accompanied by pertinent and relevant questions. Additionally, it entails a cognitive challenge aimed at fostering higher-order skills, such as knowledge application, decision making, and problem solving. Through authentic assessments, students showcase understanding, retrieve prior knowledge, establish connections between theories and practice, formulate solutions, draw conclusions, and delineate subsequent steps or actions. Lastly, authentic assessment involves evaluative judgment, encouraging students to establish criteria and standards for assessing their own performance, thereby promoting self-regulated learning.

In brief, authentic assessment surpasses traditional methods by being multifaceted and dynamic [17,35]. Grounded in multiple criteria, it focuses on students' progress toward mastery, presenting realistic, contextualized, and complex intellectual challenges.

Unlike fragmented tasks, authentic assessment identifies strengths rather than serving as a punitive measure.

2.3. Constructive Alignment

There is a need to articulate the ideas of experiential learning and authentic assessment to provide a structure to interplay with GenAI tools and integrate these into learning experiences and activities. A step forward in this direction can be found in the concept of constructive alignment [36].

If students are to engage in experiential learning for authentic assessment, teachers' fundamental task is to develop suitable learning experiences that are likely to achieve the intended learning outcomes (ILOs) [36–38]. Constructive alignment is based on three central elements: (i) intended learning outcomes (ILOs), (ii) teaching and learning activities (TLAs), and (iii) assessment tasks (ATs). Alignment is achieved by ensuring ILOs reflect the desired learning outcomes, while TLAs facilitate achieving those outcomes, and ATs assess students' attainment of the ILOs. Teaching and learning activities become crucial in accomplishing the requisite ILO verbs, whereas the AT challenge is to show evidence of students' achievements authentically.

Constructive alignment suggests that meaningful learning occurs when students actively construct knowledge and meaning through coherent, authentic, and contextualized experiences. By aligning ILOs, TLAs, and ATs, educators can promote higher-order thinking, deep understanding, and acquiring relevant skills. Overall, this view can help integrate experiential learning and authentic assessment.

This proposition helps to identify the possible interplay of GenAI tools within a structure of constructive alignment in which learning technology enhances pedagogies and extends learning environments [39]. Hence, GenAI tools can be linked to navigate "what to learn", as defined by ILOs, support "how to learn" according to the experiential learning cycle in TLAs, and effectively construct responses, execute tasks, or create products as indicated by ATs in "how to assess learning".

3. Materials and Methods

To address the research questions, this study employs an ethnographic approach to explore the nuances of integrating GenAI into experiential learning for authentic assessment. By treating GenAI tools as active subjects in the research process, the investigation aims to uncover novel possibilities in a still ill-explored field. By leveraging their text processing and generation capabilities, GenAI tools' responses will help us identify innovative alternatives for incorporating GenAI in learning activities.

Ethnographic research considers a cultural lens to the study of people's lives within their multiple formal and informal communities with the aim of observing and analyzing how people interact with each other and with their environment [40]. Ethnography seeks to access the "native's point of view" through the meanings and reports of people. Accordingly, ethnography focuses on interpretation, understanding, and representation. This is a social constructionist research approach in which several descriptions, or versions, of "reality" are considered to provide an authentic description of the world. Therefore, an ethnographic research process involves data collection from different perspectives that represent participants in their own terms and description writing followed by analysis and interpretation [40,41].

In this way, this work considers GenAI tools not merely as objects, but as study subjects, because of their capacity to provide human-like responses and seamlessly engage in conversations with humans [23]. For instance, previous works in artificial intelligence (AI) show the possibility of interviewing ChatGPT for information elicitation or obtaining insights into diverse research topics [42–45].

Using a GenAI tool as a subject that co-performs daily practices with users and impacts their interactions in which it is embedded allows people to engage and converse with it. As humans, we shape objects, and objects shape and transform our practices reciprocally [46]. Acknowledging this dynamic interaction between people and objects calls for approaches that give both entities recognizable roles in human practices and present new ways of defining and solving problems collectively with things, whose skills and functions are different from those of humans. This perspective challenges anthropocentric assumptions about the world and opens new ways of understanding objects, people, and use practices [21,47].

A plausible method for this purpose can be found in thing ethnography to aid in exploring GenAI's interaction and participation in human activities, social and cultural dimensions, and their impact on society (and education) from their perspective-to give voices to the voiceless. Accordingly, thing ethnography refers to collecting and interpreting things' perspectives from everyday data and trajectories that things provide access to and the theory-based analysis that humans undertake to identify patterns and gain novel insights into their socio-material interactions [21]. Thing ethnography involves stepping into things' shoes to explore the acting out of things' attributes, as in role-playing, to portray and empathize with the elusive "inner life of things" [48]. Cameras, microphones, and sensors are used for thing-centered data collection; however, other alternatives like interviews can be used for this purpose. Generally, ethnographic interviews are conducted within a specific social location and represent an experience with moments of interaction with declarative content [49]. In this sense, *interviews with things*, in thing ethnography, can help to access and illustrate the first-person things' subjectivities and agency by looking at the particular qualities and contexts of their experience [48]. Nevertheless, interviews with things focus on the richness of insights and inspiration generated in the process rather than on the reliability and representativeness of results.

Previous thing ethnography interventions have used different objects and sensors to obtain the perspective of things; however, given chatbots' unique text generation capacities, this work interviews GenAI tools as subjects using written conversations [23].

Currently, some of the largest, most common, and widely used online GenAI conversational chatbots available for text generation are Open AI's ChatGPT 3.5, Google Bard, Microsoft New Bing (version 96.0.1054.43 recently rebranded as Microsoft Copilot), and Anthropic's Claude 1.0. These chatbots are highly regarded because of their usage simplicity, free access, availability, and performance, despite their current limitations such as hallucinations, limited or no access to internet databases, and poor or no use of references [50,51]. Therefore, these four chatbots were selected as interview subjects within this work's ethnography methodology to obtain their perspectives and help answer the research questions. The interview subjects and their URL links are listed in Table 1 below.

Table 1. Interview subjects.

GenAI Tools	Developer	URL
ChatGPT 3.5	OpenAI	https://chat.openai.com (accessed on 16 September 2023)
Claude 1.0	Anthropic	https://genai.works/app/claude (accessed on 16 September 2023)
New Bing/Copilot Microsoft	https://www.bing.com/new (accessed on 16 September 2023);	
	https://copilot.microsoft.com/	
Bard	Google	https://bard.google.com (accessed on 16 September 2023)

The process of interviewing GenAI tools was threefold [48,52]: (i) data collection, (ii) data organization, and (iii) data analysis and interpretation. In the data collection stage, GenAI tools were interviewed on a one-to-one basis using a set of questions and themes to gather data from the perspective of the thing. This study adopted an exploratory semistructured interview to allow the interviewees to unveil the richness of their views [53] with a degree of structure and flexibility [54]. An interview guide was prepared beforehand regarding experiential learning, authentic assessment, and constructive alignment, in line with the research questions, to direct the interviewees' responses, giving transparency and reliability to the interviewing process (see Appendix A). Nevertheless, probes were included as part of a non-standardized semi-structured interviewing process to provide flexibility to the interaction with the GenAI tools and allow for further exploration of significant responses deemed for the research topic [55]. Probing is adopted to seek clarification for unclear words or phrases, complete stories the interviewer thought were unfinished, and encourage engagement with the interviewee [56].

The second stage involved becoming familiar with the collected data set and preparing and organizing it for their analysis and reporting. This stage in this work consisted of a thematic analysis driven by the supported theoretical approach and the research questions (see Section 1), which deductively informed about the necessary codes and themes for collating and grouping data [57]. Hence, GenAI tools' text answers were reviewed to understand and empathize with the thing's perspective, and they were prepared/collated in formats to categorize data systematically and to identify and report similarities, coincidences, or patterns that helped to provide a compelling argument concerning the supporting evidence.

Finally, in the last stage, the researchers interpreted data by taking an immersive exploration into the inner perspective of things and making sense of it through pedagogical theories. The researchers linked, at this stage, the analysis of the first-person view of things to the research questions to provide answers and create insights that can shed light on the relationships between objects and human practices and present new ways of defining and solving problems collaboratively with things [46]. This view refers to expanding the alternatives for understanding the possible use of GenAI tools to enhance experiential learning for authentic assessment and their integration into learning experiences in HE.

As for ethnographic studies [58–61], this work addresses the research criteria of validity by proposing a thing ethnography methodology that acknowledges the animistic nature of GenAI tools. This approach considers the access to their perspectives and reaches consistent interpretations when comparing interview results with the existing body of literature. Moreover, reliability is addressed by providing a step-by-step methodology that consistently allows for the subsequent collection, usage, and reporting of data from different chatbots. It also considers data accuracy in comparing deviations in prompts (of different GenAI tools) of the various research studies [62]. Regarding transferability, as generalizability, this research study's findings might not apply to other contexts, situations, times, and populations, as GenAI tools' views can differ from other subjects in their opinions (for example, they might not answer the same) as well as in their contextual conditions or circumstances, leading to different results and interpretations [63]. This limitation requires further data collection and validation, as in Popper's idea of falsification [59,64].

4. Results

The extracts of the interviews with ChatGPT 3.5, Google Bard, Microsoft New Bing, and Anthropic's Claude regarding the conceptual understanding of the GenAI tools and the research questions are presented here. The interview extracts were summarized in tables, and emerging themes were identified accordingly.

Questions involving the fundamental concepts guiding this work were prompted to the four selected GenAI tools to test their "understanding" of the topics. Despite these questions and answers not being a central part of this work, they illustrate the capability of the tools to build sensible and well-informed responses.

4.1. Conceptual Clarification: GenAI Responses Regarding Experiential Learning, Authentic Assessment, and Constructive Alignment

GenAI responses to the questions regarding (i) experiential learning, (ii) authentic assessment, (iii) constructive alignment, and (iv) Kolb's cycle contribution to authentic assessment are as follows.

The answers of the four GenAI tools to the question concerning Kolb's experiential learning (What is Kolb's experiential learning about?) provide a standard and complementary definition and descriptions of the stages of the learning cycle as the main themes. Further descriptions of experiential learning regarding definitions and the learning cycle stages were provided by the GenAI tools in the interviews.

Similar and complementary answers were obtained for the following question regarding the assessment of learning outcomes in real-world scenarios by effectively applying knowledge and skills: what is Wiggins' idea of authentic assessment about in learning and teaching activities? The main emerging themes in this case were the authentic assessment concept, principles, examples, and benefits. A summary of the responses to the question, including further details on authentic assessment principles, was provided during the interviews.

The four GenAI tools provided consistent and complementary descriptions regarding the following question: what is Biggs and Tang's notion of constructive alignment concerning the pedagogical design of learning and teaching activities? They all pointed to the effective alignment of ILOs, TLAs, and ATs. The identified emerging themes consisted of the constructive alignment concept, a description of the key components, and benefits.

Finally, the four GenAI tools also consistently answered the following question: how can the use of Kolb's experiential learning cycle contribute to authentic assessment? Answers were obtained regarding the four stages of the experiential learning cycle and their contribution to authentic assessment, including descriptions, activities, contributions, and examples as the main emerging themes.

Overall, the four GenAI tools provided answers that match the existing scholarly literature on the referred topics, considering definitions, characteristics, components or elements, benefits, and examples. Hence, all GenAI tools interviewed offered appropriate and reliable descriptions.

4.2. The Interplay of GenAI Tools and Experiential Learning for Authentic Assessment

Regarding the following question, the emerging themes referred to AI-enriched ILO formulation and definition through the integration of GenAI tools: how can GenAI tools be used for the formulation of intended learning outcomes within the framework of Biggs and Tang's constructive alignment? As this question is more open to interpretation, the interviewees' (i.e., GenAIs) answers covered different issues. The most completed response was elaborated by ChatGPT 3.5, which indicated that "[b]y leveraging GenAI tools in the formulation of intended learning outcomes, educators can streamline the process, promote alignment with teaching and assessment strategies, and ensure that the learning outcomes are clear, measurable, and conducive to meaningful learning experiences". A summary of the themes and suggestions is presented in Table 2 concerning the integration alternatives for ILO formulation.

Some practical, relevant instances provided by Gen AI tools are listed below to offer practical guidance in their use:

- ChatGPT 3.5: Incorporating higher-order thinking.
 - "Scenario: An educator aims to foster higher-order thinking skills in their course".
 - GenAI assistance: The educator uses the GenAI tool to generate ILOs that align with Bloom's Taxonomy levels, ensuring that students engage in critical thinking and analysis. For example: "Evaluate the ethical implications of X" or "Analyze and synthesize competing theories in the field of Y".
- Microsoft New Bing: "[...] Content Generation: Let's say one of the learning outcomes for a computer science course is "Students should be able to understand and explain the concept of binary search trees". A GenAI tool could generate a detailed explanation of binary search trees, including their structure, operations, and use cases. This content could then be used as part of the course materials [...]".
- Anthropic Claude: "Translate outcomes into student-friendly language—Make learning outcomes clearer for students using simplification, analogies, and examples".

AI-Enriched ILO Formulation	GenAI Tools	Integration Alternatives
Content generation	ChatGPT 3.5 and Microsoft New Bing	Offer diverse sets of content outcomes for consideration related to a specific topic or subject area
Summarize standards, competencies, and goals from curriculum documentation	Anthropic Claude 1.0	Digest diverse reference materials and identify key learning aims
Alignment suggestions	ChatGPT 3.5	Analyze ILOs and provide alignment suggestions
Customization for diverse learners	ChatGPT 3.5	Create customized ILOs, using input parameters, tailored to specific courses or learner groups
Multidisciplinarity	ChatGPT 3.5	Integrate and bridge different subject areas and competencies from various domains
Language refinement	ChatGPT 3.5 and Anthropic Claude 1.0	Refine the language of ILOs to make them more precise, measurable, student-friendly, and aligned with assessment criteria
Diversity	ChatGPT 3.5	Provide a broad range of ILOs, ensuring that the learning outcomes cover various cognitive levels and different aspects of learning
Examples and templates	ChatGPT 3.5	Provide examples and templates for ILOs, making it easier to create clear and effective learning outcomes
Assessment-driven ILOs	ChatGPT 3.5	Help educators create ILOs that are closely aligned with the chosen assessment tools and criteria
Feedback and iteration	ChatGPT 3.5	Provide feedback on ILOs, suggesting improvements and offering insights into alignment issues
Alignment with real-world applications	ChatGPT 3.5	Emphasize the application of knowledge in authentic contexts to increase learning relevance
Adaptation to learners' needs	ChatGPT 3.5	Dynamically adjust ILOs based on individual learner profiles, meeting their specific needs and abilities

Table 2. The use of GenAI tools in the formulation of ILOs for constructive alignment *.

* Obtained from extracts of diverse answers provided by GenAI tools during their interviews.

Two tools (i.e., ChatGPT 3.5 and Microsoft New Bing) provided sophisticated answers consisting of a hypothetical learning space and well-intended TLAs.

Moreover, emerging themes found in the responses to the question regarding integration (how can GenAI tools be integrated into teaching and learning activities while taking into account Kolb's experiential learning cycle and Biggs and Tang's constructive alignment?) refer to suggested AI-enhanced TLAs through integrating GenAI into diverse, active pedagogical approaches and experiential learning in line with ILOs. Again, ChatGPT 3.5 provided the most comprehensive answer, claiming that "GenAI tools can [...] assist educators in designing personalized, aligned, and engaging learning experiences that cater to individual learning styles while promoting reflective practice, real-world application, and continuous improvement in line with the principles of these two educational frameworks". A summary of the themes and examples can be found in Table 3 regarding the integration of GenAI tools into TLAs.

Some relevant examples of how GenAI tools can be applied at various stages of the teaching and learning process are described as follows:

- ChatGPT 3.5: Dynamic problem-solving challenges.
 - "Kolb's Influence: To encourage active experimentation and provide students with dynamic problem-solving challenges that evolve as they make decisions. GenAI can help create adaptive problem scenarios.
 - Constructive Alignment: Make sure that the problem-solving challenges align with the ILOs and assessment methods. AI can adjust the challenges based on the desired learning outcomes and provide feedback on students' problem-solving strategies".

- Google Bard: "A science teacher might use a GenAI tool to generate a realistic simulation of a scientific experiment. This would allow students to have a concrete experience of the experiment and then reflect on their results. The teacher could then use the GenAI tool to generate concept maps or other visualizations to help students understand the abstract concepts involved in the experiment".
- Anthropic Claude: Business—Marketing.
 - a. "Student teams develop a social media marketing campaign for a product launch (collaborative project).
 - b. An AI reviews their initial draft and provides feedback on how well their ideas align with marketing best practices and campaign objectives (constructive alignment).
 - c. Students post campaign ideas to a simulated social platform and analyze user reactions (active experimentation).
 - d. An AI bot plays the role of target users responding to their posts and ideas".

Table 3. The integration of GenAI tools into TLA considering experiential learning and constructive alignment *.

AI-Enhanced TLAs	GenAI Tools	GenAI Tools Integration	
Realistic case studies	ChatGPT 3.5 and Microsoft New Bing	Help create challenging scenarios as concrete experiences for reflective and hands-on learning	
Simulations or virtual labs for active experimentation	ChatGPT 3.5, Microsoft New Bing, and Anthropic Claude 1.0	 Design and customize simulations to match desired learning outcomes Apply students' knowledge and skills in a controlled, interactive environment GenAI plays a role in simulations, responding dynamically based on student actions 	
Personalized learning pathways	ChatGPT 3.5, Google Bard, Microsoft New Bing, and Anthropic Claude 1.0	Recommend individualized content and activities that align with each student's progress through the experiential learning cycle	
Storytelling and narrative learning	ChatGPT 3.5 and Anthropic Claude 1.0	Create narrative-driven learning activities or scenarios that immerse students in the subject matter as concrete experiences	
Dynamic problem-solving challenges that evolve	ChatGPT 3.5	• Create adaptive problem scenarios as students make decisions to encourage active experimentation. Adjust challenges to desired learning outcomes and provide feedback on students' problem-solving strategies	
Interactive group discussions on complex topics	ChatGPT 3.5 and Google Bard	 Facilitate these discussions by providing discussion prompts and relevant resources. Assist in generating discussion questions that encourage critical thinking and reflection 	
Multimedia-rich learning resources	ChatGPT 3.5	Suggest diverse multimedia resources that complement the learning objectives at different stages of Kolb's cycle	
Interactive gamification learning activities	ChatGPT 3.5	Design interactive gamified learning activities that offer students concrete experiences and challenges to solve within a game-based environment	
Collaborative projects ChatGPT 3.5, Google Bard, and Anthropic Claude 1.0		 Assist teams by providing research resources, suggesting project milestones, and facilitating collaboration Help teams reflect on their progress and apply course concepts to practical project tasks 	

AI-Enhanced TLAs	GenAI Tools	GenAI Tools Integration	
Adaptive prompts, questions, or quizzes	ChatGPT 3.5, Microsoft New Bing, and Anthropic Claude 1.0	 Adjust the difficulty level of questions based on the student's progress through the learning cycle and alignment with course objectives Promote active experimentation to test knowledge, promote reflection, and receive feedback 	
Virtual field trips and tours	ChatGPT 3.5	Virtual immersive experiences related to course materials, enhanced with additional information, interactive elements, and reflection prompts	
Peer review and feedback	ChatGPT 3.5	Facilitate the peer-review process by providing guidelines and facilitating reflection, experimentation, and the exchange of feedback among students	
Interactive simulations to experiment with abstract concepts	ChatGPT 3.5	Provide hints and explanations within the simulations to help students understand and apply these concepts	
Adaptive reading lists	ChatGPT 3.5	Recommend readings based on students' progress and preferences	
Multimodal learning pathways	ChatGPT 3.5	Suggest multimedia resources to enhance engagement based on students' preferences and alignment with learning objectives	
Real-time feedback	Google Bard	Provide real-time feedback for student work improvement	
Supporting blended learning	Google Bard	Provide access to online resources that supplement face-to-face instruction	
Facilitating lifelong learning	Google Bard	 Provide access to resources and opportunities to learn new things Create a personalized learning dashboard that tracks progress and recommends new learning opportunities 	
Conceptual model development Anthropic Claude 1.0		 Assist in creating models to represent key course concepts Review models and give feedback on accuracy, completeness, and areas needing improvement 	

Table 3. Cont.

* Obtained from extracts of diverse answers provided by GenAI tools during their interviews.

In this case, Anthropic Claude provided one of the most elaborated trains of ideas by providing a TLA, with the point of intervention where a GenAI could contribute to supporting the T&L process.

The emerging themes established by the responses to the question regarding the employment of GenAI tools (how can GenAI tools be employed to facilitate Wiggins' authentic assessment methods while considering Biggs and Tang's constructive alignment?) refer to suggested AI-enabled assessment methods involving higher-order skills; active, hands-on tasks; enriched rubrics; and real-world-like scenarios. ChatGPT 3.5 indicated that "[b]y integrating GenAI tools into the authentic assessment process, educators can create assessments that align with the ILOs and teaching strategies and leverage AI's capabilities for automation, personalization, and feedback generation. This approach ensures that assessment methods effectively measure students' ability to apply their learning in real-world contexts [...]". A summary of the integration of GenAI tools for assessment methods is presented in Table 4.

Generating real-world problem-solving scenarios ChatGPT 3.5, Microsoft New Bing, and Anthropic Claude 1.0 - Create authentic, complex problem-solving, and critical thinking skills Generate scenarios that require solving, and critical thinking skills Provide students with access to case studies, data, and simulations Automated peer review with Al-assisted feedback ChatGPT 3.5 and Google Bard - Implement Al-supported peer review systems that streamline the peer assessment process for constructive comments Adaptive scenario-based assessments ChatGPT 3.5 Develop adaptive scenario-based assessments where Al adjusts the scenarios and questions based on students' responses Al-enhanced portfolio assessment ChatGPT 3.5, Google Bard and Anthropic Claude 1.0 - Helps budnets organize their work, provide feedback, and track their progress Simulations and interactive virtual labs with Al feedback ChatGPT 3.5 and Google Bard - - - - Examines student protfolios for evidence of skill development and growth over students can conduct experiments Natural language processing (NLP) for easay evaluation ChatGPT 3.5 and Google Bard -	AI-Enabled Assessment Methods	GenAI Tools	GenAI Tools Integration
Automated peer review with Al-assisted feedback ChatGPT 3.5 and Coogle Bard Implement Al-supported peer review systems that streamline the peer assessment process Adaptive scenario-based assessments ChatGPT 3.5 Assist in providing feedback templates and suggestions for constructive comments Ad-enhanced portfolio assessment ChatGPT 3.5 Develop adptive scenario-based assessments where AI adjusts the scenario based assessment work, provide feedback, and track their progress Al-enhanced portfolio assessment ChatGPT 3.5, Google Bard and Anthropic Claude 1.0 ChatGPT 3.5 and Google Bard Simulations and interactive virtual labs with Al feedback ChatGPT 3.5 and Google Bard Create virtual labs where students can conduct experiments Natural language processing (NLP) for essay evaluation ChatGPT 3.5 and Google Bard Implement NLP-powered tools to evaluate essays Al-generated project challenges ChatGPT 3.5 and Google Bard Implement NLP-powered tools to evaluate essays Al-generated project challenges ChatGPT 3.5 and Google Bard ChatGPT 3.5 and Google Bard Al-generated project challenges ChatGPT 3.5 and Google Bard ChatGPT 3.5 and Anthropic Claude 1.0 Adaptive quizzes with himmediate feedback for formative self-assessment ChatGPT 3.5 and Anthropic Claude 1.0 Create adplive quizzes with Al that adjust question difficulty based on students' alignment with the 1LOs Guide how students' alignment with the 1	Generating real-world problem-solving scenarios	ChatGPT 3.5, Microsoft New Bing, and Anthropic Claude 1.0	 Create authentic, complex problem-solving scenarios that align with the ILOs Generate scenarios that require students to apply their knowledge, problem-solving, and critical thinking skills Provide students with access to case studies, data, and simulations
Adaptive scenario-based assessmentsChatGPT 3.5Develop adaptive scenario-based assessments where AI adjusts the scenarios and questions based on students responsesAI-enhanced portfolio assessmentChatGPT 3.5, Google Bard and Anthropic Claude 1.0•Help students organize their work, provide feedback, and track their progress Examines students ortsponsesSimulations and interactive virtual labs with AI feedbackChatGPT 3.5 and Google Bard•Create virtual labs where students can conduct experiments •Natural language processing (NLP) for essay evaluationChatGPT 3.5•Create virtual labs where students can conduct experimentsAl-generated project challengesChatGPT 3.5 and Google Bard•Implement NLP-powered tools to evaluate essays •Analyze the alignment with the ILOs and helping students refine their practical skills•Implement NLP-powered tools to evaluate essays •Al-generated project challengesChatGPT 3.5 and Google Bard•Create and provide feedback on the clarity and coherence of students' argumentsAdaptive quizzes with immediate feedback for formative self-assessmentsChatGPT 3.5 and Google Bard•Create adaptive quizzes with AI that adjust question and Anthropic Claude 1.0Scenario-based role-play assessmentsChatGPT 3.5 and Anthropic Claude 1.0•Create adaptive quizzes with AI that adjust question and provide feedback for and provide feedback for and provide feedback for care adaptive quizzes with AI that adjust question and fanthropic Claude 1.0Adaptive quizzes with immediate feedback for formative self-assessment	Automated peer review with AI-assisted feedback	ChatGPT 3.5 and Google Bard	 Implement AI-supported peer review systems that streamline the peer assessment process Assist in providing feedback templates and suggestions for constructive comments
AL-enhanced portfolio assessment ChatGPT 3.5, Google Bard, and Anthropic Claude 1.0 Help students organize their work, provide feedback, and track their progress. Simulations and interactive virtual labs with AI feedback ChatGPT 3.5 and Coogle Bard Crate virtual labs where students can conduct experiments. Simulations and interactive virtual labs with AI feedback ChatGPT 3.5 and Coogle Bard Crate virtual labs where students can conduct experiments. Natural language processing (NLP) for essay evaluation ChatGPT 3.5 and Coogle Bard Implement NLP-powered tools to evaluate essays. Al-generated project challenges ChatGPT 3.5 and Google Bard Implement NLP-powered tools to evaluate essays. Al-generated project challenges ChatGPT 3.5 and Google Bard Implement NLP-powered tools to evaluate essays. Al-generated project challenges ChatGPT 3.5 and Google Bard Implement NLP-powered tools to evaluate essays. Al-generated project challenges ChatGPT 3.5 and Google Bard Implement NLP-powered tools to evaluate essays. Adaptive quizzes with immediate feedback for formative self-assessment ChatGPT 3.5 and Anthropic Claude 1.0 Implement NLP-powered tools to evaluate essays. Scenario-based role-play assessments ChatGPT 3.5 and Anthropic Claude 1.0 Generate role-play scenarios that align with the ILOs Generate role-play scenarios that align with the ILOs Generate role-play scenarios that align with the ILOs G	Adaptive scenario-based assessments	ChatGPT 3.5	Develop adaptive scenario-based assessments where AI adjusts the scenarios and questions based on students' responses
Simulations and interactive virtual labs with AI feedbackChatGPT 3.5 and Google BardCreate virtual labs where students can conduct experimentsSimulations and interactive virtual labs with AI feedbackChatGPT 3.5 and Google BardProvide real-time feedback on their actions, ensuring alignment with the ILOs and helping students refine their practical skillsNatural language processing (NLP) for essay evaluationChatGPT 3.5Implement NLP-powered tools to evaluate essays Analyze the alignment of the essay content with the ILOs and provide feedback on the clarity and coherence of students' argumentsAI-generated project challengesChatGPT 3.5 and Google BardGenerate project challenges that align with the ILOs Suggest project topics, requirements, and criteria, ensuring that they align with the intended learning outcomesAdaptive quizzes with immediate feedback for formative self-assessmentChatGPT 3.5 and Anthropic Claude 1.0Create adaptive quizzes with AI that adjust question difficulty based on students' alignment with the ILOs Guide how students' alignment with the ILOsScenario-based role-play assessmentsChatGPT 3.5Generate role-play scenarios that align with the ILOs 	AI-enhanced portfolio assessment	ChatGPT 3.5, Google Bard, and Anthropic Claude 1.0	 Help students organize their work, provide feedback, and track their progress Examines student portfolios for evidence of skill development and growth over time, mapping artefacts to intended learning outcomes
Natural language processing (NLP) for essay evaluationChatGPT 3.5Implement NLP-powered tools to evaluate essays Analyze the alignment of the essay content with the ILOs and provide feedback on the clarity and coherence 	Simulations and interactive virtual labs with AI feedback	ChatGPT 3.5 and Google Bard	 Create virtual labs where students can conduct experiments Provide real-time feedback on their actions, ensuring alignment with the ILOs and helping students refine their practical skills
AI-generated project challengesChatGPT 3.5 and Google BardGenerate project challenges that align with the ILOs Suggest project topics, requirements, and criteria, ensuring that they align with the intended 	Natural language processing (NLP) for essay evaluation	ChatGPT 3.5	 Implement NLP-powered tools to evaluate essays Analyze the alignment of the essay content with the ILOs and provide feedback on the clarity and coherence of students' arguments
Adaptive quizzes with immediate feedback for formative self-assessmentChatGPT 3.5 and Anthropic Claude 1.0Create adaptive quizzes with AI that adjust question 	AI-generated project challenges	ChatGPT 3.5 and Google Bard	 Generate project challenges that align with the ILOs Suggest project topics, requirements, and criteria, ensuring that they align with the intended learning outcomes Help students collaborate on projects, gather data, and create presentations
Scenario-based role-play assessmentsChatGPT 3.5• Generate role-play scenarios that align with the ILOs • Guide how students' performance aligns with the assessment criteria and the constructive 	Adaptive quizzes with immediate feedback for formative self-assessment	ChatGPT 3.5 and Anthropic Claude 1.0	 Create adaptive quizzes with AI that adjust question difficulty based on students' alignment with the ILOs Provide immediate feedback for each question, aligning with the assessment criteria
Creating assessment rubricsMicrosoft New Bing and Anthropic Claude 1.0Create rubrics for learning outcome achievementsCompetency-based assessmentAnthropic Claude 1.0Review student work products, like reports, designs, presentations, etc., and provide feedback on how well they demonstrate mastery of core competencies for the field or professionAutomated scoringAnthropic Claude 1.0Assist by automating routine scoring while teachers focus on higher-order evaluation and feedback	Scenario-based role-play assessments	ChatGPT 3.5	 Generate role-play scenarios that align with the ILOs Guide how students' performance aligns with the assessment criteria and the constructive alignment framework
Competency-based assessmentAnthropic Claude 1.0Review student work products, like reports, designs, presentations, etc., and provide feedback on how well they demonstrate mastery of core competencies for the field or professionAutomated scoringAnthropic Claude 1.0Assist by automating routine scoring while teachers focus on 	Creating assessment rubrics	Microsoft New Bing and Anthropic Claude 1.0	Create rubrics for learning outcome achievements
Automated scoring Anthropic Claude 1.0 Assist by automating routine scoring while teachers focus on higher-order evaluation and feedback	Competency-based assessment	Anthropic Claude 1.0	Review student work products, like reports, designs, presentations, etc., and provide feedback on how well they demonstrate mastery of core competencies for the field or profession
0	Automated scoring	Anthropic Claude 1.0	Assist by automating routine scoring while teachers focus on higher-order evaluation and feedback

Table 4. The use of GenAI tools to facilitate authentic assessment considering constructive alignment *.

extracts of diverse answers provided by GenAI tools durin ng their interviews ned ror

Some practical examples offered by GenAI tools concerning the proposals in Table 4 are listed below:

- Chat GPT 3.5: Adaptive scenario-based assessments.
 - "Biggs and Tang's Constructive Alignment: Create assessments that adapt to students' progress and align with the ILOs.
 - GenAI Facilitation: Develop adaptive scenario-based assessments where AI adjusts the scenarios and questions based on students' responses. AI ensures that each student's assessment experience is tailored to their alignment with the ILOs".
- Google Bard: "A business teacher might use a GenAI tool to generate a realistic simulation of a job interview. This could be used to assess students' communication and interview skills".
- Microsoft New Bing: "Creating Rubrics: For a course on public speaking with a learning outcome of "Students should be able to deliver a persuasive speech", a GenAI tool could create a rubric that assesses various aspects of public speaking, such as clarity of speech, strength of argument, and audience engagement".
- Anthropic Claude: History class.
 - a. "Students participate in a roleplay simulation acting as historic figures (performance-based).
 - b. An AI assesses their ability to accurately portray the figures based on provided profiles (competency-based).
 - c. Students self-reflect on their decisions and strategies in character (formative self-assessment). The AI reviews reflections and provides feedback".

In this case, all of the GenAI tools provided refined responses that were helpful to answer this challenging question.

4.3. The Use of GenAI Tools in Learning Activities for Authentic Assessment

Responses to the question regarding alternatives to integrate GenAI tools (what different alternatives can be identified to integrate GenAI tools into the learning activities associated with each of the four stages of Kolb's experiential learning cycle, all while aligning with the principles of authentic assessment proposed by Biggs and Tang?) involve themes concerning integrating GenAI tools into each experiential learning stage with specific AI-enhanced activities that promote concrete experiences, reflection, conceptualization, and practical learning. According to ChatGPT 3.5, integrating GenAI tools can enhance experiential learning while adhering to the principles of authentic assessment, namely, alignment with real-world contexts, reflection and critical thinking, practical application, and personalization and feedback. A summary is presented in Table 5 concerning the integration of GenAI tools into AI-enhanced learning activities at each stage of the experiential learning cycle.

Two integrated examples, provided by GenAI tools, are presented below to guide their implementation:

- ChatGPT 3.5:
 - Concrete experience (CE): "AI-Enhanced Virtual Reality (VR) Experiences: Create immersive VR experiences using GenAI tools that allow students to explore historically significant places, scientific simulations, or cultural events. Students can interact with the VR environment to gain concrete experiences".
 - Reflective observation (RO): "AI-Powered Reflective Journaling: Implement AI-powered journaling platforms that help students reflect on their experiences. AI can provide prompts based on their concrete experiences, guiding them to deeper reflection".

- Abstract conceptualization (AC): "AI-Driven Concept Mapping: Employ AIdriven concept mapping tools that assist students in organizing and synthesizing their abstract conceptualizations. These maps can help students clarify their understanding and connect concepts, aligning with authentic assessment by demonstrating knowledge construction".
- Active experimentation (AE): "AI-Adaptive Decision-Making Simulations: Create decision-making simulations using AI that challenge students to actively experiment with various strategies. The AI can adjust the scenarios based on students' decisions, providing a dynamic and aligned learning experience".
- Anthropic Claude: Business course—negotiation skills:
 - a. Students roleplay a business negotiation against an AI bot playing the negotiation partner (concrete experience).
 - b. The AI assesses the negotiation strategy and adapts its responses to drive reflection (reflective observation).
 - c. Students write a report applying negotiation theory to analyze the experience (abstract conceptualization).
 - d. The AI reviews the report based on the rubric criteria tied to the learning outcomes (authentic assessment).

The previous two examples from ChatGPT 3.5 and Anthropic Claude show the ability of GenAI tools to provide functional answers to complex questions concerning T&L challenges.

Experiential Learning Stage	GenAI Tools	AI-Enhanced Learning Activities	GenAI Tools Integration
Concrete experience	ChatGPT 3.5 Microsoft New Bing, Google Bard, and Anthropic Claude 1.0	AI-generated scenario-based simulations	 Create realistic simulations that immerse students in authentic scenarios related to the course content AI dynamically adjusts parameters and situations in an immersive virtual environment in response to student actions
	ChatGPT 3.5 and Google Bard	AI-enhanced virtual field trips	Create enhanced virtual field trips or tours by providing interactive elements and real-time information
	ChatGPT 3.5	AI-enhanced virtual reality (VR) experiences	Create immersive VR experiences that allow students to historically explore significant places, scientific simulations, or cultural events
	ChatGPT 3.5 and Bard	AI-generated scenario challenges	Generate complex, real-world scenarios or problems that simulate challenges faced in specific professions or industries for problem solving
	Google Bard	Gamified learning	Create gamified learning experiences that make learning fun and engaging
	Anthropic Claude 1.0	AI adaptive tutoring and chat box	Provide personalized guidance and questioning in simulations and AI roleplaying

Table 5. The use of GenAI tools to facilitate authentic assessment considering constructive alignment*.

Experiential Learning Stage	GenAI Tools	AI-Enhanced Learning Activities	GenAI Tools Integration
Reflective observation	ChatGPT 3.5	AI-powered discussion forums	 Engage students in reflective discussions on their concrete experiences Assist in moderating discussions, summarizing key points, and providing prompts for deeper reflection
	ChatGPT 3.5, Microsoft New Bing, and Anthropic Claude	AI-generated reflection prompts	Generate personalized reflection prompts based on students' experiences to think critically about their experiences and promote self-reflection
	ChatGPT 3.5, Google Bard, and Anthropic Claude	AI-powered reflective journaling	Help students reflect on their experiences, guiding them to deeper reflection
	ChatGPT 3.5 and Google Bard	Automated peer reflection facilitation	Facilitate peer reflection by grouping students and generating reflection questions or discussion topics based on their shared concrete experiences
	Google Bard and Anthropic Claude	Self-assessment	Create self-assessment tools that listen, ask follow-up questions, and help students track their progress and identify areas where they need to improve
Abstract conceptualization	ChatGPT 3.5 and Microsoft New Bing	AI-generated conceptual exercises	Create abstract conceptualization exercises that challenge students to connect their concrete experiences to theoretical concepts by providing hints and explanations
	ChatGPT 3.5	AI-personalized conceptual quizzes	Generate personalized quiz questions to align students' prior concrete experiences with the abstract concepts they have encountered, providing an assessment of their understanding
	ChatGPT 3.5 and Google Bard	AI-driven concept mapping	Employ AI-driven concept mapping tools that assist students in organizing and synthesizing their abstract conceptualizations to clarify their understanding
	ChatGPT 3.5	AI-generated conceptual analysis tasks	Provide data or scenarios for analysis to apply abstract concepts to real-world problems
	Google Bard	AI mnemonic devices	GenAI tools can be used to create mnemonic devices that help students remember important information
	Google Bard	AI modeling	GenAI tools can be used to create models that help students understand abstract concepts
Active experimentation	ChatGPT 3.5 and Google Bard	AI-enhanced project-based learning	Provide real-world project suggestions, learning resources, and automated feedback
	ChatGPT 3.5 and Google Bard	AI-adaptive decision-making simulations	 Create decision-making simulations to actively experiment with various strategies Adjust scenarios based on students' decisions, providing a dynamic and aligned learning experience

Table 5. Cont.

Experiential Learning Stage	GenAI Tools	AI-Enhanced Learning Activities	GenAI Tools Integration
Active experimentation	ChatGPT 3.5	AI-simulated experiment design	 Create AI-driven experiment design platforms to execute virtual experiments in a controlled environment. Provide real-time feedback on experiment design and outcomes
	ChatGPT 3.5	AI-enhanced project collaboration	Suggest project milestones, identify potential project risks, and help students actively experiment with project management strategies
	Google Bard	Internships	Connect students with internship opportunities to gain real-world experience in their field
	Microsoft New Bing	AI decision-making scenarios	Create new scenarios for decision making where students can apply their knowledge and validate their learning

Table 5. Cont.

* Obtained from extracts of diverse answers provided by GenAI tools during their interviews.

5. Discussion

To answer the research questions, a discussion in this section delves into an exploration of experiential learning, authentic assessment, and constructive alignment; the interplay of GenAI tools with experiential learning for authentic assessment; and alternatives for integrating these tools in teaching and learning practice. Additionally, limitations and future work are examined to delimit the contributions of this work.

5.1. Conceptual Clarification of Experiential Learning, Authentic Assessment, and Constructive Alignment as Explained by GenAI Tools

The interpretations provided by the four GenAI tools to increasingly complex questions show a common understanding of Kolb's experiential learning. Their level of abstraction recognizes the cyclical nature of the learning process; however, the four tools present different depths in the theoretical construct and detail in the implementation. In summary, the four tools can offer practical experiences complemented by reflection, abstract conceptualization, and active experimentation.

The explanations of authentic assessment provided by the GenAI tools demonstrate a shared understanding of its principles and purpose, but also with different levels of detail and exemplification. These explanations emphasize the importance of assessing students' abilities in real-world contexts, promoting meaningful learning and preparing students for future challenges. Moreover, they highlight the need for its practical implementation in educational settings for curriculum design, assessment design, and the role of teachers in facilitating authentic assessment.

Regarding constructive alignment, the different interpretations of GenAI tools also reveal a shared understanding of its principles and significance in pedagogical design. Constructive alignment is a valuable framework for creating coherent, aligned, studentcentered learning experiences to construct knowledge through students' active engagement. Again, different and complementary levels of detail and examples were provided.

Lastly, concerning integrating experiential learning into authentic assessment, the shared interpretations of the GenAI tools refer to comprehensive learning, student engagement, and learning relevance by incorporating real-world experiences and activities aligned with the learning cycle. Additionally, there is an emphasis on reflective practice and the acquisition of real-world skills, which prepares students for their future careers and lifelong learning. The iterative nature of both the cycle and authentic assessment supports

continuous feedback and improvement, ensuring that students can continually refine their understanding and skills.

Summing up, GenAI tools can help understand pedagogical theories by providing clear definitions, explanations, and practical applications from diverse perspectives. Educators, researchers, and learners can leverage these insights to create more meaningful and effective student-centered learning experiences. However, each GenAI tool offers unique perspectives, depth of understanding, and strengths. While ChatGPT 3.5 mainly provides comprehensive explanations, including definitions, principles/elements, and practical guidance, Google Bard emphasizes practical aspects and real-world applications, New Bing focuses on clarifying theories and concepts, and Anthropic Claude highlights goals and principles.

5.2. Findings on the Interplay of GenAI Tools and Experiential Learning for Authentic Assessment

The relationship between GenAI tools, authentic assessment, and experiential learning has been explored according to the structure of constructive alignment as an alternative to guide and articulate their integration effectively. Accordingly, this discussion is threefold, covering the constructive alignment structure of ILOs, TLAs, and ATs in the following subsections.

5.2.1. The Use of GenAI Tools in the Formulation of ILOs for Constructive Alignment

Referring to the interview results in Table 2, integrating GenAI tools into the formulation and definition of ILOs offers support in content generation, customization, alignment, and language refinement, promoting personalized and learner-centric education. This view provides the possibility of improving the quality and pertinence of ILOs. Moreover, these tools facilitate ILO formulation by enriching their multidisciplinarity, diversity, and the application of knowledge in real-world contexts, all of which are essential aspects of constructive alignment. ILO feedback and iterative improvements are also encouraged, fostering ongoing quality enhancement in education.

The differences in the proposed integration of GenAI tools for ILO formulation mainly revolve around the tools' unique functions and strengths. For instance, ChatGPT 3.5 excels in most proposals, especially in context generation, customization, and adaptability; New Bing focuses on context generation; while Anthropic Claude emphasizes the analysis of existing content. Google Bard does not provide a particular distinctive answer on this topic. On the other hand, similarities include content generation, alignment support, language refinement, and adaptation to learners' needs. However, they all provide various possibilities for integrating GenAI tools for ILO formulation (see Table 2).

5.2.2. The Integration of GenAI Tools into TLA Considering Experiential Learning and Constructive Alignment

Integrating GenAI tools into TLA offers alternatives to designing AI-enhanced activities and resources that align with the different phases of the experiential learning cycle and ensure alignment with learning outcomes (see Table 3). Additionally, they facilitate personalization, engagement, and real-time feedback, promoting a dynamic and learner-centric approach to education. In summary, the differences among these GenAI tools primarily relate to the scope of their functions and areas of specialization. ChatGPT 3.5 offers a wide range of capabilities, including simulations, gamification, real-time feedback, and more, while other tools may have more specific functions. In terms of exemplification, ChatGPT 3.5 and Anthropic Claude provide the most precise and explicit practical examples to guide the tools' implementation. However, similarities include personalization and adaptation, narrative learning, collaborative projects, feedback and assessment, the recommendation of multimedia resources, support for blended learning, and the facilitation of lifelong learning. These tools offer diverse capabilities to enhance TLAs within an experiential learning framework for constructive alignment.

5.2.3. The Use of GenAI Tools to Facilitate Authentic Assessment Considering Constructive Alignment

Integrating GenAI tools contributes diverse options to creating authentic, relevant, and effective assessments, emphasizing alignment with intended learning outcomes, real-world applications, ongoing improvement, and student-centered learning. The responses of GenAI tools to integrating authentic assessment methods within the constructive alignment framework exhibit differences and similarities. ChatGPT 3.5 and Anthropic Claude focus on creating realistic problem-solving scenarios that align with ILOs. ChatGPT 3.5 and Google Bard contribute to AI-assisted peer review, streamlining the feedback process. Furthermore, ChatGPT 3.5 and Google Bard emphasize creating adaptive assessments based on students' responses. On the other hand, Microsoft New Bing and Anthropic Claude concentrate on competency-based assessment, automating routine scoring, and developing assessment rubrics. Anthropic Claude specifically engages in reviewing student work products for core competencies.

Regarding AI-enhanced portfolio assessment and simulations with AI feedback, Chat-GPT 3.5 and Google Bard share common ground, focusing on monitoring students' progress over time. While all of these GenAI tools work towards authentic assessment, each one has a unique role. ChatGPT 3.5 is more versatile, contributing to various assessment methods. Google Bard, on the contrary, is more concentrated, mainly proposing real-time feedback and gamification activities. Microsoft New Bing and Anthropic Claude are more focused on competency-based assessments and rubric development. Nevertheless, referring to the tools' exemplification, ChatGPT 3.5 and Anthropic Claude provide sound, practical examples to guide their integration.

These differences in emphasis reflect the varying strengths and capabilities of the GenAI tools, but their collective contribution seeks to align assessment practices more closely with constructive alignment principles.

Accordingly, concerning RQ1, a relationship can be established between GenAI tools and experiential learning for authentic assessment following the structure of constructive alignment as follows:

- GenAI tools can enrich ILO formulation, enhance their quality and pertinence, and validate their definition, clarity, and content. GenAI tools provide a wide range of possibilities for integrating GenAI tools into ILO formulation (see Table 2 for integration alternatives).
- GenAI tools can help TLAs to develop AI-enhanced activities and resources that align with experiential learning and other pedagogies to ensure alignment with learning outcomes, opening the gate for authentic assessments (see Table 3 for activity examples).
- GenAI tools provide diverse options for creating AI-enabled/assisted assessments that underscore alignment with ILOs, real-world or contrived applications, ongoing improvement, and student-centered learning (see Table 4 for authentic assessment options).
- Overall, GenAI tools can articulate with (and integrate into) experiential learning for authentic assessment through an AI-supported coherent structure of ILOs, TLAs, and ATs. In this case, GenAI tools can act as agents-to-define what to learn, how to learn, and how to assess learning, supporting and facilitating the instructional/pedagogical design of learning experiences. Additionally, these tools also offer action-oriented possibilities to become agents-to-teach-and-learn-with and agents-to-assess-learning-with. Therefore, GenAI tools can become transformative resources to support teaching and learning roles in teaching practice, learning activities, and within learning environments. This view calls for the design of pedagogical interventions in which GenAI tools are purposively integrated to achieve specific teaching and learning aims and goals.

5.3. Findings on the Use of GenAI Tools in Specific Learning Activities for Authentic Assessment

The interview results provide insights into how GenAI tools can transform learning activities across different stages of the experiential learning process, from concrete experience to abstract conceptualization and active experimentation (see Table 5). These tools have the potential to foster deeper learning through reflective and critical thinking and practical skill development. Hence, GenAI tools can be effectively integrated into AI-enabled experiential learning to create more dynamic and enriching educational experiences while considering authentic assessment.

In examining the proposals by ChatGPT 3.5, Google Bard, Microsoft New Bing, and Anthropic Claude, it becomes evident that these AI-driven tools exhibit distinct specializations and areas of focus. ChatGPT 3.5 stands out as a versatile instrument, spanning multiple stages of experiential learning and offering a broad spectrum of functionalities, such as generating simulations and facilitating discussions. In contrast, Google Bard specializes in constructing intricate scenarios and gamified experiences, emphasizing problem solving and engagement, primarily during the early phases of the experiential learning cycle. New Bing takes a unique role by mainly concentrating on generating scenarios for decision making, which plays a vital role in the active experimentation stage, aligning with practical applications within educational contexts. Claude primarily focuses on personalized guidance, roleplaying, and facilitating discussions within simulations, catering to the human interaction aspect of learning, particularly in the earlier stages. Again, in terms of exemplification, ChatGPT 3.5 and Anthropic Claude provide the most transparent and explicit practical examples to guide the tools' incorporation. Despite these differences, there are striking similarities among these tools, most notably concerning their commitment to personalization, interactive learning elements, feedback and reflection mechanisms, dynamic adaptation, real-world application, support for collaborative learning, and integrating assessment and evaluation features. These can collectively enhance experiential learning for authentic assessment across diverse educational settings.

Concerning RQ2, alternatives were proposed for including GenAI tools in learning activities considering experiential learning and authentic assessment:

- GenAI tools can support AI-enhanced activities across each stage of the experiential learning cycle. GenAI tools are also linked to integrating diverse, active pedagogical approaches and strategies such as adaptive learning, project-based learning, learning challenges, internships, field trips, collaborative learning, journaling, and gamification. They also cover individual, group, independent, or supervised activities for learning outcome development. Additionally, AI-enhanced activities also point to decision making, problem solving, modeling, and simulations, which allow for the development of high-level cognitive skills in real-world or contrived scenarios. Therefore, GenAI tools offer integrative pedagogical approaches and strategies within experiential learning activities for the authentic assessment of ILOs.
- GenAI tools might be regarded as *agents-to-learn-with*. They actively interact with learners as AI-enabled participants in their undertakings to accomplish their ILOs, provide support and feedback, and genuinely assess their accomplishments. This view demands the design of pedagogical interventions to directly support learners and their interactions with GenAI tools to improve their learning experiences and achievements.

5.4. Limitations

This work encompasses methodological and GenAI-related limitations, but also ethical concerns. In methodological terms, thing ethnography gives a voice to objects because of their supposed animistic nature. As GenAI tools can reply to prompts in human-like language, limitations and biases may exist regarding their consideration as subjects and capacity to articulate valid responses in thing interviewing. GenAI responses require accuracy and relevance verification by cross-referencing these with authoritative sources. In addition, they may only sometimes capture the full range of opinions and approaches within a given field. Another limitation is the potential variability in responses from the same GenAI tools over time. This constraint shares similarities with human responses in social research, necessitating careful crafting of the research process to attain stability in reports and responses by following a falsification process [64]. The results show that the GenAI responses

concerning the fundamental notions were consistent with those in the existing body of literature, but differences in detail, scope, and exemplification existed. Nevertheless, these results positively contribute to this work's research validity and transferability. However, responses regarding the research questions require further investigation through empirical work and practical cases in real-world educational scenarios.

Something that none of the GenAI tools provided was a feedback strategy to evaluate the effectiveness of the recommendations after their implementation. This lack of feedback makes it difficult to confirm the level of success or failure of any of the GenAI proposals. This absence results from how the interviews were conducted and the need for more explicit questions on this issue. Therefore, further research on GenAI tools' ability to act as reflective subjects/observers needs to be conducted.

Referring to GenAI tools, it is essential to use them as supplementary resources and apply their insights in a responsible, human-centered, and context-specific manner, considering that human thinking and expertise are still inimitable. Accordingly, GenAI tools must be a source of information and feedback to continuously test and validate the insights gained from their use, obtain real-world reactions from students and colleagues, and stay updated on the latest developments in AI.

In ethical terms, this work highlights the potential limitations and concerns regarding GenAI accessibility and inclusivity for all possible users, restrained human interactions and intellectual development, hidden bias, the potential manipulation of generated content, data privacy and security, transparency usage, and the purposes AI serves. These ideas call for a solid reflective teaching practice to anticipate the possible risks. There is also an ethical link to the responsible use of GenAI tools regarding information quality and usage, which must always adhere to human interpretation, judgment, and decision making. GenAI tools must not be used to make pedagogical human decisions, but should be used as supportive resources.

5.5. Future Work

Future work points to (i) the validation and testing of the proposed GenAI integration options for experiential learning activities and authentic assessment, (ii) application to specific educational real-world contexts, and (iii) creating clear guidelines for addressing limitations and ethical concerns.

A research agenda can be created to evaluate the proposed GenAI tools' integration options and their impacts on learning effectiveness and authentic assessment. This proposition can also include assessing their use to enrich ILOs, TLAs, and ATs. However, this view could extend to other active pedagogical approaches, such as challenge-based learning, gamification, or project-based learning. Concerning experiential learning, the proposed AI-enhanced activities at each stage of the learning cycle can be further explored to detail tasks in specific learning environments, scenarios, and learning disciplines. There is also the need to assess the direct contribution these tools can make to students' learning, including knowledge growth and skill development and the enrichment/enhancement of their learning experiences.

Future work is also required to exemplify the use of these tools in real-world contexts to create application cases for research, discussion, and dissemination purposes. Integrating GenAI tools into teaching and learning activities is promising but must be clarified. Therefore, practical implementations are required to delve into the grassroots of the use and possible implications. Also, new potential applications or limitations could emerge and be recognized from their use.

Finally, ethical concerns exist regarding the use of this technology beyond academic integrity, intellectual property, and plagiarism; it is necessary to explore new ways of addressing these issues regarding GenAI tools' reliability, responsible use, accessibility and inclusion, privacy and security, and transparency. This view calls for instrumenting preventive guidelines and actionable plans for appropriate and ethical usage.

6. Conclusions

In conclusion, this research has shed light on the interplay between GenAI tools and experiential learning for authentic assessment. By integrating GenAI tools into the constructive alignment framework for the instructional design of intended learning outcomes (ILOs), teaching and learning activities, and assessment tasks, we provided valuable guidance and practical examples to answer the research questions. Our work has not only identified alternatives for incorporating GenAI tools into learning activities while considering experiential learning and authentic assessment, but it has also opened exciting possibilities for enriching the human–GenAI interaction.

Our contributions extend beyond using GenAI tools as mere *agents-to-write* or *agents-to-answer* questions to become *agents-to-support* learning experiences. We highlighted their potential to act as *agents-to-think-about* ILOs and real-world complex learning scenarios, *agents-to-teach-and-learn-with* to facilitate active learning experiences, *agents-to-assess-learning-with* authentic assessment tasks, and *agents-to-learn-with* experiential learning activities for authentic assessment. These findings underscore the transformative role of GenAI tools in enhancing teaching and learning efficacy and effectiveness.

However, it is crucial to emphasize that the responsible use of GenAI tools is paramount. We must recognize the need to understand how to use GenAI effectively, ensuring that precise questions are asked to prevent GenAI hallucinations and the generation of fake content. Moreover, ethical implications and risks associated with GenAI tools should be carefully considered, and human agency and learning needs should always take precedence.

While our research provides a foundation for understanding the potential of GenAI tools in education, we acknowledge that further practical work and implementation in real-world educational settings are necessary to validate their contributions to learning. This ongoing exploration of GenAI's role in education is critical for accomplishing learning objectives and outcomes, making the appropriate use of these tools while integrating pedagogical approaches tailored to learners' needs. In this evolving landscape, the responsible use of GenAI is central to supporting human interactions and the broader purpose of education.

Author Contributions: Conceptualization, D.E.S.-N.; methodology, D.E.S.-N., E.V.-P. and R.M.-V.; validation, E.V.-P., R.M.-V. and L.M.; formal analysis, D.E.S.-N.; investigation, D.E.S.-N., E.V.-P. and R.M.-V.; data curation, D.E.S.-N.; writing—original draft preparation, D.E.S.-N.; writing—review and editing, D.E.S.-N., E.V.-P, R.M.-V. and L.M.; visualization, D.E.S.-N. and L.M.; supervision, D.E.S.-N.; project administration, D.E.S.-N.; funding acquisition, L.M. All authors have read and agreed to the published version of the manuscript.

Funding: The authors would like to acknowledge the financial support of Writing Lab, Institute for the Future of Education, Tecnologico de Monterrey, Mexico, in the production of this work.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data supporting this study's findings are available from the corresponding author.

Conflicts of Interest: The authors declare no conflicts of interest.

Appendix A Interview Questions

A literature-enriched interview guide was elaborated to conduct the interaction with GenAI tools during the methodology's data collection stage as follows:

- 1. Conceptual clarification.
 - What is Kolb's experiential learning about?
 - What is Biggs and Tang's notion of constructive alignment concerning the pedagogical design of learning and teaching activities?

- What is Wiggins' idea of authentic assessment about in learning and teaching activities?
- How can the use of Kolb's experiential learning cycle contribute to authentic assessment?
- 2. The interplay of GenAI tools and experiential learning for authentic assessment (RQ1).
 - How can GenAI tools be used for the formulation of intended learning outcomes within the framework of Biggs and Tang's constructive alignment? Provide examples.
 - How can GenAI tools be integrated into teaching and learning activities while taking into account Kolb's experiential learning cycle and Biggs and Tang's constructive alignment? Provide examples.
 - How can GenAI tools be employed to facilitate Wiggins' authentic assessment methods while considering Biggs and Tang's constructive alignment? Provide examples.
- 3. The use of GenAI tools in specific learning activities for authentic assessment (RQ2).
 - What different alternatives can be identified to integrate GenAI tools into the learning activities associated with each of the four stages of Kolb's experiential learning cycle, all while aligning with the principles of authentic assessment proposed by Biggs and Tang? Provide additional alternatives and examples.

References

- 1. Euchner, J. Generative AI. Res.-Technol. Manag. 2023, 66, 71–74. [CrossRef]
- 2. Jovanovic, M.; Campbell, M. Generative Artificial Intelligence: Trends and Prospects. Computer 2022, 55, 107–112. [CrossRef]
- 3. Stokel-Walker, C.; Van Noorden, R. What ChatGPT and Generative AI Mean for Science. Nature 2023, 614, 214–216. [CrossRef]
- van der Zant, T.; Kouw, M.; Schomaker, L. Generative Artificial Intelligence. In *Philosophy and Theory of Artificial Intelligence*; Studies in Applied Philosophy, Epistemology and Rational Ethics; Springer: Berlin/Heidelberg, Germany, 2013; pp. 107–120.
 [CrossRef]
- 5. Eke, D.O. ChatGPT and the Rise of Generative AI: Threat to Academic Integrity? J. Responsible Technol. 2023, 13, 100060. [CrossRef]
- Chinonso, O.E.; Theresa, A.M.-E.; Aduke, T.C. ChatGPT for Teaching, Learning and Research: Prospects and Challenges. *Glob. Acad. J. Humanit. Soc. Sci.* 2023, *5*, 33–40. [CrossRef]
- 7. UNESCO; Holmes, W.; Miao, F. Guidance for Generative AI in Education and Research; UNESCO: Paris, France, 2023; ISBN 978-92-3-100612-8.
- 8. Geerling, W.; Mateer, G.D.; Wooten, J.; Damodaran, N. ChatGPT Has Aced the Test of Understanding in College Economics: Now What? *Am. Econ.* 2023, *68*, 233–245. [CrossRef]
- Albert, D.; Li, T. Insights from Teaching with AI: How ChatGPT Can Enhance Experiential Learning and Assist Instructors. SSRN Electron. J. 2023, 1–24. [CrossRef]
- 10. Fuchs, K. Exploring the Opportunities and Challenges of NLP Models in Higher Education: Is Chat GPT a Blessing or a Curse? *Front. Educ.* **2023**, *8*, 1166682. [CrossRef]
- Smith, A.; Hachen, S.; Schleifer, R.; Bhugra, D.; Buadze, A.; Liebrenz, M. Old Dog, New Tricks? Exploring the Potential Functionalities of ChatGPT in Supporting Educational Methods in Social Psychiatry. *Int. J. Soc. Psychiatry* 2023, 69, 1882–1889. [CrossRef]
- 12. Kokoç, M.; Altun, A. Effects of Learner Interaction with Learning Dashboards on Academic Performance in an E-Learning Environment. *Behav. Inf. Technol.* **2019**, *40*, 161–175. [CrossRef]
- Tsai, C.-W.; Shen, P.-D.; Tsai, M.-C. Developing an Appropriate Design of Blended Learning with Web-Enabled Self-Regulated Learning to Enhance Students' Learning and Thoughts Regarding Online Learning. *Behav. Inf. Technol.* 2011, 30, 261–271. [CrossRef]
- Martínez-Cerdá, J.-F.; Torrent-Sellens, J.; González-González, I. Promoting Collaborative Skills in Online University: Comparing Effects of Games, Mixed Reality, Social Media, and Other Tools for ICT-Supported Pedagogical Practices. *Behav. Inf. Technol.* 2018, 37, 1055–1071. [CrossRef]
- 15. Hernández-Lara, A.B.; Serradell-López, E. Student Interactions in Online Discussion Forums: Their Perception on Learning with Business Simulation Games. *Behav. Inf. Technol.* **2018**, *37*, 419–429. [CrossRef]
- 16. Crawford, J.; Cowling, M.; Allen, K.-A. Leadership Is Needed for Ethical ChatGPT: Character, Assessment, and Learning Using Artificial Intelligence (AI). *J. Univ. Teach. Learn. Pract.* 2023, 20, 1–20. [CrossRef]
- 17. Wiggins, G. The Case for Authentic Assessment. Pract. Assess. Res. Eval. 2019, 2, 2. [CrossRef]
- 18. Kolb, D.A. *Experiential Learning: Experience as the Source of Learning and Development;* Prentice-Hall: Englewood Cliffs, NJ, USA, 1984; ISBN 978-0-13-295261-3.

- 19. dos Santos, R. Enhancing Chemistry Learning with ChatGPT and Bing Chat as Agents to Think with: A Comparative Case Study. SSRN Electron. J. 2023, 1–27. [CrossRef]
- 20. Lawrie, G. Establishing a Delicate Balance in the Relationship between Artificial Intelligence and Authentic Assessment in Student Learning. *Chem. Educ. Res. Pract.* 2023, 24, 392–393. [CrossRef]
- Giaccardi, E.; Cila, N.; Speed, C.; Caldwell, M. Thing Ethnography: Doing Design Research with Non-Humans. In Proceedings of the 2016 ACM Conference on Designing Interactive Systems, Brisbane, Australia, 4–8 June 2016; pp. 377–387.
- Yurman, P.; Reddy, A.V. Drawing Conversations Mediated by AI. In Proceedings of the Creativity and Cognition, Venice, Italy, 20–23 June 2022; pp. 56–70.
- 23. Michel-Villarreal, R.; Vilalta-Perdomo, E.; Salinas-Navarro, D.E.; Thierry-Aguilera, R.; Gerardou, F.S. Challenges and Opportunities of Generative AI for Higher Education as Explained by ChatGPT. *Educ. Sci.* 2023, *13*, 856. [CrossRef]
- 24. Benkert, C.; van Dam, N. Experiential Learning: What's Missing in Most Change Programs. Available online: https://www. mckinsey.com/capabilities/operations/our-insights/experiential-learning-whats-missing-in-most-change-programs (accessed on 15 December 2023).
- Salinas-Navarro, D.E.; Garay-Rondero, C.L.; Rodriguez Calvo, E.Z. Experiential Learning Spaces for Industrial Engineering Education. In Proceedings of the 2019 IEEE Frontiers in Education Conference (FIE), Covington, KY, USA, 16–19 October 2019; pp. 1–9.
- Kolb, A.Y.; Kolb, D.A. Learning Styles and Learning Spaces: Enhancing Experiential Learning in Higher Education. *Acad. Manag. Learn. Educ.* 2005, *4*, 193–212. [CrossRef]
- Bradberry, L.A.; De Maio, J. Learning By Doing: The Long-Term Impact of Experiential Learning Programs on Student Success. J. Political Sci. Educ. 2019, 15, 94–111. [CrossRef]
- Healey, M.; Jenkins, A. Kolb's Experiential Learning Theory and Its Application in Geography in Higher Education. J. Geogr. 2000, 99, 185–195. [CrossRef]
- Freeman, S.; Eddy, S.L.; McDonough, M.; Smith, M.K.; Okoroafor, N.; Jordt, H.; Wenderoth, M.P. Active Learning Increases Student Performance in Science, Engineering, and Mathematics. *Proc. Natl. Acad. Sci. USA* 2014, 111, 8410–8415. [CrossRef] [PubMed]
- 30. Lalley, J.P.; Miller, R.H. The Learning Pyramid: Does It Point Teachers in the Right Direction? Education 2007, 128, 64–79.
- 31. Wiggins, G. A True Test: Toward More Authentic and Equitable Assessment. Phi Delta Kappan 2011, 92, 81–93. [CrossRef]
- 32. Koh, K.H. Authentic Assessment. In Oxford Research Encyclopedia of Education; Oxford University Press: Oxford, UK, 2017.
- 33. Villarroel, V.; Bloxham, S.; Bruna, D.; Bruna, C.; Herrera-Seda, C. Authentic Assessment: Creating a Blueprint for Course Design. *Assess. Eval. High. Educ.* **2018**, *43*, 840–854. [CrossRef]
- 34. Villarroel, V.; Boud, D.; Bloxham, S.; Bruna, D.; Bruna, C. Using Principles of Authentic Assessment to Redesign Written Examinations and Tests. *Innov. Educ. Teach. Int.* **2019**, *57*, 38–49. [CrossRef]
- Merrett, C.G.; Adams, J.; Johnson, A.W.; Swenson, J.E.S. Collaborating with Aviation Museums to Enhance Authentic Assessments for Aerospace Structures. In Proceedings of the ASEE Annual Conference and Exposition, Conference Proceedings 2023, Baltimore, MD, USA, 25 June 2023.
- 36. Biggs, J. Enhancing Teaching through Constructive Alignment. High. Educ. 1996, 32, 347–364. [CrossRef]
- 37. Biggs, J. Aligning Teaching and Assessing to Course Objectives. In Proceedings of the Teaching and Learning in Higher Education: New Trends and Innovations, Aveiro, Portugal, 13–17 April 2003.
- Biggs, J.B.; Tang, C.S. *Teaching for Quality Learning at University: What the Student Does*, 4th ed.; SRHE and Open University Press Imprint; McGraw-Hill, Society for Research into Higher Education & Open University Press: Maidenhead, UK; New York, NY, USA, 2011; ISBN 978-0-335-24275-7.
- 39. Radcliffe, D. A Pedagogy-Space-Technology Framework for Designing and Evaluating Learning Places. In *Learning Spaces in Higher Education: Positive Outcomes by Design*; The University of Queensland: Brisbane, Australia, 2019; pp. 9–16.
- 40. Eriksson, P.; Kovalainen, A. Qualitative Methods in Business Research, 2nd ed.; SAGE: Los Angeles, CA, USA, 2016; ISBN 978-1-4462-7339-5.
- 41. Reeves, S.; Kuper, A.; Hodges, B.D. Qualitative Research Methodologies: Ethnography. BMJ 2008, 337, a1020. [CrossRef]
- 42. Asch, D.A. An Interview with ChatGPT About Health Care. NEJM Catal. 2023, 4, 1–8.
- 43. Iskender, A. Holy or Unholy? Interview with Open AI's ChatGPT. Eur. J. Tour. Res. 2023, 34, 3414. [CrossRef]
- 44. Karakose, T.; Demirkol, M.; Aslan, N.; Köse, H.; Yirci, R. A Conversation with ChatGPT about the Impact of the COVID-19 Pandemic on Education: Comparative Review Based on Human–AI Collaboration. *Educ. Process Int. J.* 2023, 12, 7–25. [CrossRef]
- 45. Tülübaş, T.; Demirkol, M.; Özdemir, T.Y.; Polat, H.; Karakose, T.; Yirci, R. An Interview with ChatGPT on Emergency Remote Teaching: A Comparative Analysis Based on Human–AI Collaboration. *Educ. Process Int. J.* **2023**, *12*, 93–110. [CrossRef]
- 46. Giaccardi, E.; Speed, C.; Cila, N.; Caldwell, M.L. Things as Co-Ethnographers: Implications of a Thing Perspective for Design and Anthropology. In *Design Anthropological Futures*; Routledge: London, UK, 2020; pp. 235–248. [CrossRef]
- Cila, N.; Giaccardi, E.; Tynan-O'Mahony, F.; Speed, C.; Caldwell, M. Thing-Centered Narratives: A Study of Object Personas. In Proceedings of the 3rd Seminar Research Network for Design Anthropology, Aarhus, Denmark, 22–23 January 2015.
- Chang, W.-W.; Giaccardi, E.; Chen, L.-L.; Liang, R.-H. Interview with Things: A First-thing Perspective to Understand the Scooter's Everyday Socio-material Network in Taiwan. In Proceedings of the 2017 Conference on Designing Interactive Systems, Edinburgh, UK, 10–14 June 2017.

- 49. Rinaldo, R.; Guhin, J. How and Why Interviews Work: Ethnographic Interviews and Meso-Level Public Culture. *Sociol. Methods Res.* **2019**, *51*, 34–67. [CrossRef]
- 50. Rudolph, J.; Tan, S.; Tan, S. War of the Chatbots: Bard, Bing Chat, ChatGPT, Ernie and beyond. The New AI Gold Rush and Its Impact on Higher Education. *JALT* 2023, *6*, 364–389. [CrossRef]
- Mollick, E. How to Use AI to Do Practical Stuff: A New Guide. Available online: https://www.oneusefulthing.org/p/how-to-use-ai-to-do-practical-stuff (accessed on 15 December 2023).
- 52. Huang, Y.-C.; Cheng, Y.-T.; Liang, R.-H.; Hsu, J.Y.; Chen, L.-L. Thing Constellation Visualizer: Exploring Emergent Relationships of Everyday Objects. *Proc. ACM Hum.-Comput. Interact.* 2021, *5*, 479. [CrossRef]
- 53. Bell, E.; Bryman, A.; Harley, B. (Eds.) Business Research Methods; Oxford University Press: Oxford, UK, 2022; ISBN 978-0-19-886944-3.
- 54. Merriam, S.B. *Qualitative Research and Case Study Applications in Education;* A Joint Publication of the Jossey-Bass Education Series and the Jossey-Bass Higher and Adult Education Series; Rev. and Expanded; Jossey-Bass Publishers: San Francisco, CA, USA, 1998; ISBN 978-0-7879-1009-9.
- 55. Saunders, M.; Lewis, P.; Thornhill, A. *Research Methods for Business Students*, 7th ed.; Always Learning; Pearson: Harlow, UK; Munich, Germany, 2016; ISBN 978-1-292-01662-7.
- 56. King, N.; Horrocks, C.; Brooks, J.M. *Interviews in Qualitative Research*, 2nd ed.; SAGE: Los Angeles, CA, USA; London, UK; New Delhi, India; Singapore; Washington, DC, USA; Melbourne, Australia, 2019; ISBN 978-1-4462-7497-2.
- 57. Braun, V.; Clarke, V. Using Thematic Analysis in Psychology. Qual. Res. Psychol. 2006, 3, 77–101. [CrossRef]
- 58. LeCompte, M.D.; Goetz, J.P. Problems of Reliability and Validity in Ethnographic Research. *Rev. Educ. Res.* **1982**, *52*, 31–60. [CrossRef]
- 59. de Zeeuw, G. *Three Phases of Science: A Methodological Exploration;* Centre for Systems and Information Sciences, University of Humberside: Lincoln, UK, 1996.
- 60. Vahl, M. Doing Research in the Social Domain. In *Systems for Sustainability;* Springer: Boston, MA, USA, 1997; pp. 147–152. [CrossRef]
- 61. Leung, L. Validity, Reliability, and Generalizability in Qualitative Research. J. Fam. Med. Prim Care 2015, 4, 324–327. [CrossRef]
- 62. Eisenhardt, K.M. Building Theories from Case Study Research. Acad. Manag. Rev. 1989, 14, 532. [CrossRef]
- 63. Willis, E.M. The Problem of Time in Ethnographic Health Care Research. Qual. Health Res. 2010, 20, 556–564. [CrossRef]
- 64. Popper, K.R. The Logic of Scientific Discovery; Routledge Classics; Repr. 2008 (Twice); Routledge: London, UK, 2008; ISBN 978-0-415-27843-0.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.